Author’s response to reviews

Title: Clozapine and desmethylclozapine: correlation with neutrophils and leucocytes counting in Mexican patients with schizophrenia

Authors:

Mayela Vaquero-Baez (auroramayela@yahoo.com)

Araceli Díaz-Ruíz (adiaz@innn.edu.mx)

Luis Tristán-López (carbolit2001@yahoo.com.mx)

Carlos Aviña-Cervantes (karloz2@hotmail.com)

Carlos Torner (ctorner@correo.xoc.uam.mx)

Jesús Ramírez-Bermúdez (jesusramirezb@yahoo.com.mx)

Sergio Montes (montesergio@yahoo.com)

Camilo Rios (crios@correo.xoc.uam.mx)

Version: 2 Date: 09 Sep 2019

Author’s response to reviews:

Answers to the Editor

Dear Sir

We have produced a "clean version" of the manuscript.

Authors CT and JRB both participated in the design of the study and the final analysis of the results. Both are clinicians (psychiatrists) and pharmacologists, thus, their participation was substantial during the writing of the manuscript and they answered some of the questions raised by the reviewers regarding the characteristics of patients enrolled in the study. Now this is stated under "author's contributions"

Regarding funding sources, one was employed to fund the analyses of the drugs in plasma from patients (CONACYT 26172) and the other one was a fellowship to AM B-V. This was clarified in the funding section.

We merged the conflict of interests and the competing interests sections.
Sincerely,

Camilo Rios, PhD,

on behalf of all authors.